Author’s response to reviews

Title: Guillain-Barre syndrome associated with hemorrhagic fever with renal syndrome in China: a case report

Authors:
Jie Jiao (jiaojie143@163.com)
Lei Wu (wlyingsh@163.com)
Jianyuan Yin (289447919@qq.com)
Xiaojiao Quan (15248942365@sina.cn)
Wei Chen (drchenwei@vip.sina.com)
Jie Hu (hujie0325@foxmail.com)

Version: 6 Date: 19 Mar 2018

Author’s response to reviews:

1. Please clarify that the Consent for Publication form you have attached is from the patient himself. This is still unclear due to your statement still reading “Written informed consent was obtained from the patient’s daughter for publication of this case report and any accompanying images”. Please state clearly in the ‘Consent for publication’ statement that the patient provided written informed consent for publication.

The patient's name is Jianyu Liu. After I got his consent, I forgot to change the declaration in the manuscript. I've change it in the recent edition (line 205) and I am sorry for wasting your time because of my carelessness.

2. There is still significant overlap with previously published works. checked